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Minutes for the May 16, 2012, CERES  

Processing Oversight Board Meeting  
 

Meeting began at 1:00 p.m. in Building 1268C Room 2316. 

Meeting Notes 

There were no comments or questions on the minutes from the last meeting.   

Announcements 

None 

 

Open Action Items 

1. Discuss the ISCCP-D2like-FlxDay/Nit file names with Dave Doelling.  (2/16/11 – 

Jonathan) 

2. Provide answers to Raja’s questions she emailed recently.  (3/16/11 – Jonathan) 

 

FM5 Status 

We’ve received all data over the last few days despite emails about delays. 

 

VIIRS Subsetter 

We currently receive VIIRS data from IDPS, but that may change to Land PEATE.  It looks like 

we get more data from Land PEATE.  This could be a problem long term because Land 

PEATE is going away at some point.  We will receive data from IDPS and Land PEATE for 

a week or two, then one data stream will be turned off when we give the okay.   

 

We’ll be providing a subsetter code change to Goddard. 

 

SCCR Discussion (SCCRs that have been submitted or updated since the last meeting) 

CERESlib 898 –TISA Gridding updates. 

   

There were no issues identified, so this SCCR will be approved. 

 

Toolkit/GCC/SLES11 Testing  

Walt presented slides on the new OS, Toolkit, and compilers.  There are two test platforms – 

bc114 (P6) and ac09 (x86).  Testing needs to be completed by June 29.  The testing strategy 

and schedule were shown.  Tom and Denise have done some testing; no problems so far. 

 

Systems 

Files reverting back to previous version.  Thomas sent an RT ticket about this problem; Chris is 

waiting for a reply from IBM. 

 

Sharon is having problems running DX on clouds-b-blue.  She’s running out of memory.  Chris 

suggested that she try running on x86. 

 

SPG10 is schedule to be decommissioned in two months. 
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Status of Open SCCRs 

892 – No updates. 

809 – Tammy and Rita are running a month on AMI-P and AMI, respectively, with updated code 

from Sunny.  AMI-P and AMI handle NaN differently.  May 25 is the earliest possible 

delivery date. 

814 – Will deliver additional updates to CERESlib.  Will need a new CERESlib SCCR for these 

updates. 

867 – On target for June 1 delivery.  Jeremie having trouble with hdiff; Nelson is going to look 

into it. 

898 – No updates. 

888 – No updates. 

877 – BDS portion converted and being tested.  SDSs are working, but there are some issues 

with Vdata.  About 50% complete. 

878 – Will need to update AJSS for PGE CER1.1P8. 

 

Additions to the Posted Reports 

Toolkit 

No updates to status report. 

 

CERESlib 

No updates to status report. 

   

Perl_Lib 

No updates to status report. 

 

CERES Processing System 

No updates to status report. 

 

Instrument 

Working on Toolkit testing. 

  

ERBE-like 

Delivered Terra and Aqua Edition3 Spectral Response Function files. 

 

Clouds 

No updates to status report. 

 

Convolution 

Aqua MODIS Collection 6 has been run; waiting for approval from Science to release. 

 

Inversion 

No updates to status report. 

 

SARB 

Offline SYNI runs are finished.  Might be able to reduce quota back to what it used to be.  Tom 

will confirm with TISA. 
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TISA Gridding & GGEO 

No updates to status report. 

 

TISA Averaging 

No updates to status report. 

 

Production Processing Database/Automation 

Expect to get Draft Transition Plan out today. 

 

CERES Websites 

No updates to status report. 

 

CERES Ordering Tool 

New release out soon. 

 

FM5 

No updates to status report. 

 

 

Meeting ended at 2:05 p. m. 


